

Origin, Migration Routes and Worldwide Population Genetic Structure of the Wheat Yellow Rust Pathogen Puccinia striiformis f.sp. tritici

Sajid S. Ali, Pierre Gladieux, Marc Leconte, Angelique Gautier, A.F. Justesen, Mogens S. Hovmøller, Jerome Enjalbert, Claude C. Pope de Vallavieille

▶ To cite this version:

Sajid S. Ali, Pierre Gladieux, Marc Leconte, Angelique Gautier, A.F. Justesen, et al.. Origin, Migration Routes and Worldwide Population Genetic Structure of the Wheat Yellow Rust Pathogen Puccinia striiformis f.sp. tritici. PLoS Pathogens, 2014, 10 (1), pp.003903. 10.1371/journal.ppat.1003903 . hal-01019256

HAL Id: hal-01019256 https://hal.science/hal-01019256

Submitted on 27 May 2020 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Origin, Migration Routes and Worldwide Population Genetic Structure of the Wheat Yellow Rust Pathogen *Puccinia striiformis* f.sp. *tritici*

Sajid Ali^{1,2,3}*, Pierre Gladieux^{4,5}, Marc Leconte¹, Angélique Gautier¹, Annemarie F. Justesen³, Mogens S. Hovmøller³, Jérôme Enjalbert⁶, Claude de Vallavieille-Pope¹

1 INRA UR 1290 BIOGER-CPP, Thiverval-Grignon, France, 2 Institute of Biotechnology and Genetic Engineering, the University of Agriculture, Peshawar, Pakistan, 3 Department of Agroecology, Aarhus University, Slagelse, Denmark, 4 UMR 8079 Ecologie Systematique Evolution, Univ. Paris-Sud., CNRS-F, Orsay, France, 5 Department of Plant and Microbial Biology, University of California, Berkeley, Berkeley, California, United States of America, 6 INRA UMR 320 Génétique Végétale, Ferme du Moulon, Gif sur Yvette, France

Abstract

Analyses of large-scale population structure of pathogens enable the identification of migration patterns, diversity reservoirs or longevity of populations, the understanding of current evolutionary trajectories and the anticipation of future ones. This is particularly important for long-distance migrating fungal pathogens such as Puccinia striiformis f.sp. tritici (PST), capable of rapid spread to new regions and crop varieties. Although a range of recent PST invasions at continental scales are well documented, the worldwide population structure and the center of origin of the pathogen were still unknown. In this study, we used multilocus microsatellite genotyping to infer worldwide population structure of PST and the origin of new invasions based on 409 isolates representative of distribution of the fungus on six continents. Bayesian and multivariate clustering methods partitioned the set of multilocus genotypes into six distinct genetic groups associated with their geographical origin. Analyses of linkage diseguilibrium and genotypic diversity indicated a strong regional heterogeneity in levels of recombination, with clear signatures of recombination in the Himalayan (Nepal and Pakistan) and near-Himalayan regions (China) and a predominant clonal population structure in other regions. The higher genotypic diversity, recombinant population structure and high sexual reproduction ability in the Himalayan and neighboring regions suggests this area as the putative center of origin of PST. We used clustering methods and approximate Bayesian computation (ABC) to compare different competing scenarios describing ancestral relationship among ancestral populations and more recently founded populations. Our analyses confirmed the Middle East-East Africa as the most likely source of newly spreading, hightemperature-adapted strains; Europe as the source of South American, North American and Australian populations; and Mediterranean-Central Asian populations as the origin of South African populations. Although most geographic populations are not markedly affected by recent dispersal events, this study emphasizes the influence of human activities on recent long-distance spread of the pathogen.

Citation: Ali S, Gladieux P, Leconte M, Gautier A, Justesen AF, et al. (2014) Origin, Migration Routes and Worldwide Population Genetic Structure of the Wheat Yellow Rust Pathogen *Puccinia striiformis* f.sp. tritici. PLoS Pathog 10(1): e1003903. doi:10.1371/journal.ppat.1003903

Editor: Bruce A. McDonald, ETH Zurich, Switzerland

Received August 10, 2013; Accepted December 12, 2013; Published January 23, 2014

Copyright: © 2014 Ali et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by the European Integrated Project BIOEXPLOIT (FOOD-CT-2005-513959), the French project EMERFUNDIS (ANR 07-BDIV-003), the Higher Education Commission, Pakistan's Overseas scholarship Phase-II Batch-II No.1247, and Marie Curie grant FP7-PEOPLE-2010-IOF-No.273086. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: bioscientist122@yahoo.com

Introduction

Domestication of ecosystems, climate change and the expanding global trade have accelerated the pace of pathogen emergence and spread [1,2]. Widely distributed and genetically homogenous crop genotypes are conducive for rapid pathogen emergence and subsequent propagation across large areas. Even when novel pathogens are initially endemic to restricted geographical areas, they can rapidly emerge in new regions, provided they encounter a farmland with susceptible hosts and favorable environmental conditions. In widely grown food crops, many pathogens were introduced long ago, now being geographically widespread, and therefore they do not come to mind when considering invasive pathogens [3]. Although introductions could have occurred centuries ago, the evolution of such ubiquitous pathogens remains a significant cause of concern due to the risk of re-emergence caused by accidental dissemination of new, multi-virulent races [4,5] or new, highly aggressive strains [6]. An accurate understanding of the origin, distribution of diversity reservoirs and past and recent migration routes of these pathogens is crucial to understand current epidemics, develop risk-assessment models and alleviate the potential adverse effects of disease emergence [7,8]. This is particularly true for pathogens capable of long-distance migration, for which any newly advantageous mutant (increased virulence, aggressiveness or resistance to fungicides) has the potential to spread over large geographical area [9].

Yellow (stripe) rust on wheat, which is caused by *Puccinia striiformis* f.sp. *tritici* (PST), is present in most wheat-growing regions of the world [5,10,11,12,13,14]. The pathogen has major negative impacts on wheat production due to re-emergences and invasions

Author Summary

Domestication of ecosystems, climate change and expanding global trade have accelerated the pace of disease emergence, caused by their introduction into new areas with susceptible hosts or the spread of new damaging pathogen genotypes. The wheat yellow rust pathogen (PST) is a pathogen with recent reports of invasions, significantly affecting worldwide wheat production. However, its origin and ancient migration routes remained unclear and the source areas of newly spreading strains were largely unknown. This information is important to understand the trajectories of current invasions and forecast the future spread, and more generally develop risk-assessment models of pathogen emergence. We analyzed a set of worldwide representative isolates of PST, which enabled to identify six different area-specific populations. Using population genetics tools, we identified its centre of diversity in the Himalayan and near Himalayan region. We also identified the source of the recently emerged populations; Middle East-East Africa as the source of high-temperature-adapted strains spreading worldwide; Europe as the source of American and Australian populations; and Mediterranean-Central Asian populations as the origin of South African populations. We demonstrate the influence of human activities on the recent long-distance spread of the disease, though most geographic populations are not markedly affected by recent dispersal events.

[5,6,15]. As for most ubiquitous pathogens of major crops, the origin, introduction pathways and current population structure of wheat yellow rust remains largely unknown. Although an origin in Transcaucasia has been hypothesized based on disease prevalence and geographical barriers [16,17], it has never been assessed in light of new knowledge on the population structure of PST. Longdistance dispersal by wind is thought to play a key role in the dissemination of the disease. The fungus is capable of longdistance migration, with well-documented cases of recurrent reestablishment of pathogen populations in areas where there are no host plants during summer/winter to allow the pathogen survival, as for the main wheat-growing provinces of north-eastern China [9]. Such spread can be due to successive jumps from field to field by this polycyclic disease throughout the season (as in the USA [18]), as well as direct long-distance migration caused by winds, as documented between England and Denmark [19]. Accidental spore transport via human travel may also contribute to the intercontinental dispersal of the pathogen, as exemplified by the introduction of PST into Australia in 1979 from Europe through contaminated clothing or goods [15].

Despite the capacity for long-distance migration, the worldwide spread of PST is relatively recent, with most emergences reported only within the last decades. The pathogen reached South Africa in 1996 from an unknown source, but the first pathotypes detected were similar to those present in the Middle East and Mediterranean regions [6,20]. PST was first reported in South America in the early 20th century, with an unknown origin [16,21]. More recently, an expansion of the geographic range of PST into the warm climate of south-eastern USA [22] was shown to be due to the emergence of an aggressive strain adapted to higher temperatures than usually reported to be optimal for PST [23,24]. The same genotype was found in Australia two years later, while another closely related one was observed in Europe, Central and West Asia and East Africa [6]. In addition to recently colonized areas, the disease is known to periodically re-emerge through the acquisition of new virulences. These events are well documented for pathotypes carrying virulence against resistance gene $\gamma r g$, with a first report in 1986 in Eastern Africa (Ethiopia) and subsequent invasions of the Middle East, Pakistan and India, reaching Bangladesh in only 12 years [5]. The geographical origins of most of the emerging strains are unknown. The population structure of PST is therefore likely to display the hallmarks of a complex mixture of re-emergences over continuous wheat-growing areas and rare founder events due to long-distance migration. Recent spreads of the disease are likely to induce marked changes in patterns of population differentiation among regions, potentially erasing the signature of more ancient colonization events.

Several recent studies, using different genetic markers, investigated the population structure of PST over relatively large geographic scales. Analyses revealed an overall clonal population structure with low genetic diversity worldwide [15,25,26,27], except China and Pakistan [28,29]. Continental dispersal of the pathogen was well evidenced [30], leading to the conclusion that a newly emerged strain/pathotype can sweep all the worldwide populations [9,26]. However, despite the importance of this pathogen and increased research efforts, aspects of the worldwide population structure of the pathogen, both in terms of population subdivision and diversity, remain undetermined.

In the present study, we assembled a representative set of isolates from a larger collection of PST populations from the worldwide geographical range of the fungus and analyzed their genetic variability using a single set of highly variable genetic markers. Our objectives were: (i) to identify the main genetic groups in modern PST populations; (ii) to test the Himalayan region as putative centre of diversity; and (iii) to identify the geographical origin of recently emerged populations and investigate the ancestral relationships among geographically spaced populations.

Materials and Methods

Selection of isolates

A set of 409 isolates was selected to represent 11 geographical regions on six continents (Africa, Asia, Australia, Europe, North America and South America) from a collection of more than 4,000 isolates available at Institut National de la Recherche Agronomique (INRA), France and the Global Rust Reference Centre, Aarhus University, Denmark. The selection was made to maximize the representation of each population (partially assessed previously by AFLP, microsatellites and/or virulence profiles [6,10,25,28,29,31,32]), while balancing the number of isolates for each previously described genetic group, and trying to cover the geographic distribution of each. Isolates representative of aggressive strains were selected from the two recently emerged aggressive strains, PstS1 (associated with the post-2000 epidemics in the USA and Australia), and the Euro-Asian strain, PstS2, as well as a set of aggressive isolates frequently reported in Southern Europe, PstS3, which were less aggressive than PstS1 and PstS2 [24]. Isolates from the recently invading populations (e.g. South Africa) were included to infer on their source using non-parametric and STRUCTURE analyses, and were therefore not included in populationbased analyses. Details regarding the number of isolates are shown in Table 1.

Molecular genotyping

For most isolates, DNA was already available, having been previously extracted through modified CTAB protocols [19,33]. For isolates from Pakistan (2008), Nepal (2008) and China (2005), DNA was extracted from 5 mg of spores following Ali et al. [34]. All of the isolates were multiplied from single pustule lesions to avoid a mixture of genotypes. Molecular genotyping was carried **Table 1.** Sampling regions and number of isolates selected for the analysis of the worldwide population structure of wheat yellow

 rust pathogen Puccinia striiformis f.sp. tritici.

Geographical origin	Country	Number of isolates	Sampling year
Older native populations			
South-East Asia	China	71	2004–05
	Nepal	55	2008
	Pakistan	68	2004–2006, 2008
Middle-East	Afghanistan	7	2009
	Cyprus	9	2005–2006
	Iran	17	2005
	Israel	7	2005–2006
	Lebanon	5	2006
	Turkey	10	2005
	Yemen	12	2003, 2005
East Africa	Eritrea	23	2002–2005
Central Asia	Azerbaijan	11	2005
	Kazakhstan	6	2005
	Kyrgyzstan	7	2005
	Uzbekistan	1	2003
Mediterranean area	Algeria	7	2005–2006
	Italy	2	1998–2006
	Morocco	2	2006
	Portugal	4	2006
	Spain	2	2006
	Tunisia	4	2005, 2007
NW Europe	Denmark	7	1995–2002
	France	5	1997–2008
	United Kingdom	5	1975, 1978, 1991–1998
Recent invasions			
North America	Mexico	2	1989, 2002
	USA	10	1981–1983, 1991–1997
South America	Argentina	1	2010
	Brazil	6	2010
	Chile	7	2010
	Uruguay	10	2010
South Africa	South Africa	6	1996–2004
Aggressive isolates	Australia	4	2002–2004
	Denmark	7	2001–2006
	France	1	1997
	Mexico	2	2002–2003
	USA	6	2000–2005
Total		409	

doi:10.1371/journal.ppat.1003903.t001

out using a set of 20 microsatellite loci in three multiplex reactions, with subsequent separation of the PCR products using a Beckman Coulter CEQ-8000 DNA Analyzer. Electrophorograms were processed using the CEQ-8000 Genetic Analysis System Software (Beckman Coulter [34]).

Analyses of population subdivision

We investigated the existence of different genetic pools of PST using both multivariate and model-based Bayesian clustering approaches. These methods avoid the clustering of individuals on a priori knowledge such as geographical locations that may artificially group different genetic lineages introduced in the same area, hindering the detection of admixture events among them [35]. Multivariate analyses were carried out using discriminant analyses of principal components, implemented in the ADEGENET package in R environment [36]. The number of clusters was identified based on the Bayesian Information Criterion (BIC), as suggested by Jombart et al. [36].

The model-based Bayesian method implemented in STRUCTURE 2.2 [37] was mostly used to confirm results of multivariate analyses, bearing in mind that this method makes the assumption of linkage equilibrium and that violations of this hypothesis for instance due to asexual reproduction can lead to spurious assignments and overestimate the number of clusters [38]. The rationale of this method is to assign multilocus genotypes to different clusters while minimizing the Hardy-Weinberg disequilibrium and the gametic phase disequilibrium between loci within clusters (where the number of clusters may be unknown). The Monte Carlo Markov Chain (MCMC) sampling scheme was run for 200,000 iterations with a 100,000 burn-in period, with K ranging from 1 to 10 and 20 independent replications for each K. The STRUCTURE outputs were processed with CLUMPP [39]; a G'statistic greater than 80% was used to assign groups of runs to a common clustering pattern.

The relatedness among geographically spaced populations was plotted using a neighbor-joining population tree based on the genetic distance D_A [40], as implemented in the POPULATION program [41]. Significance was assessed using 1,000 bootstraps. The level of population differentiation was assessed using pairwise F_{ST} statistics among pairs of populations (GENETIX 4.05.2 [42]).

Analyses of genetic variability and recombination

The quality of the set of markers, with respect to the inference of population structure, was tested by assessing the ability of the set of microsatellite loci to detect multilocus genotypes (MLGs) under panmixia, using GENCLONE [43]. The redundancy of the set of loci was tested by estimating the linkage disequilibrium among different loci and generating 1,000 random permutations with GENETIX 4.05.2 [42]. Within-population variability was assessed using allele richness and gene diversity, calculated with FSTAT 2.9.3 [44]. Genotypic diversity was estimated with MULTILOCUS 1.3 [45]. Private allelic richness was estimated using a rarefaction approach, implemented in ADZE [46]. Observed (Ho) and unbiased expected heterozygosity (He) were computed using GENETIX 4.05.2 [42]. The null hypothesis of the Hardy-Weinberg equilibrium within each population was tested using the exact test implemented in GENEPOP 4.0 [47]. Calculations were performed both on the whole dataset and on the clone-corrected data (i.e., a dataset in which only one representative of each repeated MLG is kept). Only the clone-corrected data are reported in cases where the two datasets yielded different results because the sampling during epidemics would result in over-representation of certain clones due to the recent clonal burst at local and seasonal scale, which may bias the population genetic analyses [48].

Ancestral relationship and migration patterns among populations

We used approximate Bayesian computation (ABC) to compare different competing scenarios describing ancestral relationship among populations. The approach bypasses the calculation of exact likelihoods rendering it efficient for complex population genetic models such as those underlying biological invasions [49,50]. The rationale is to simulate datasets assuming different parameter values under different scenarios to estimate posterior probabilities of competing scenarios and the posterior distributions of the demographic parameters under a given scenario using comparisons between simulated and observed data sets based on summary statistics (e.g. genetic distance between populations). As different scenarios compared were defined based on the results of population structure analyses, they are therefore described in the results section. These scenarios globally contrasted various hypotheses concerning the source and sink relationships among the different genetic groups identified using clustering methods.

Simulations were performed using DIVABC [51], model selection and parameter estimation was carried out using the abc package in **R** [52]. A total of 5×10^5 simulated data sets was generated for each scenario under the generalized stepwise mutation model, with two parameters: the mean mutation rate (I) and the mean parameter (P) of the geometric distribution used to model the length of mutation events (in number of repeats). The mean mutation rate was drawn from a uniform distribution of 10^{-4} to 10^{-3} , while the mutation rate of each locus was drawn from a gamma distribution (mean = μ , shape = 2). The parameter P was kept in the range of 0.1 to 0.3. A range of 40 contiguous allelic states was kept for each locus, characterized by individual value of mutation rate (l_{L}) and the parameter of the geometric distribution (P_{I}) , which were obtained from Gamma distribution (with mean = 1, range 5×10^{-5} to 5×10^{-2} for I_L ; and mean = P, shape = 2, shape 0.01–0.09 for P_L). Mean number of alleles per locus, mean genetic diversity [53], mean variance in allele size, genetic differentiation between pairwise groups, F_{ST} [54], and genetic distance ($\delta \mu^2$ [55] were used as summary statistics.

Relative posterior probabilities of different scenarios were estimated by fitting a multinomial logistic regression between the summary statistics and a polychotomous variable corresponding to the different model indexes [56], using 1% of simulated dataset closest to the observed data. The posterior distributions of parameters were estimated for the most likely scenario using a local linear regression [57,58] on 1% simulated datasets closest to the observed data. Confidence in model choice was assessed using a leave-one-out method [52] for each model we drew 100 of the 5×10^{5} simulated datasets used for model selection and treated them as observed datasets (i.e., pseudo-observed datasets). Posterior probabilities of competing models were evaluated for each pseudo-observed dataset, using all remaining simulated datasets and the same methodology as described for the observed dataset. Confidence in model choice was then estimated using the number of pseudo-observed dataset that gave higher posterior probability to the model they had been simulated with. In tests of goodness-of-fit (i.e., model checking), we simulated 100 datasets of similar number of markers as observed datasets and calculated for each dataset the average across loci of several test quantities. The set of test quantities included the summary statistics used in analyses of the observed dataset. However, because using the same statistics in parameter inference and model checking can overestimate the quality of the fit [51], we selected additional summary statistics that had not been used in parameter inferences: mean allele size variance across loci, mean index of classification, and for each population pair the mean gene diversity, mean allelic richness, and mean allele size variance across loci. Test statistics computed from observed data were then ranked against the distributions obtained from simulated datasets [58].

Results

Summary of genetic variation

We performed multilocus genotyping of 409 PST isolates, representatives of a worldwide collection, using a set of 20 microsatellite markers. Plotting the multilocus genotypes detected against the number of loci re-sampled showed that the full set of SSRs was sufficient to discriminate clonal lineages (supporting information; Fig. S1). No significant linkage disequilibrium was found among SSR loci (data not shown), suggesting a lack of redundancy among markers. Some of the loci were monomorphic in certain geographical areas, except that China had no such locus



Figure 1. Discriminant analysis of principal components (DAPC) analysis of worldwide PST populations sampled from different geographical regions. The Eigen values of the analysis suggest that the first two components explained the maximum genetic structure of the dataset (A). The Bayesian information criteria (BIC) supported six distinct genetic groups (B). Scatter-plot of the worldwide distribution of PST isolates into six genetic groups (C). doi:10.1371/journal.ppat.1003903.g001

and Pakistan had only one (RJN-12; supporting information; Table S1).

Population subdivision

Genotypes clearly clustered according to their geographical origin in the non-parametric discriminant analysis of principal components (DAPC) analysis. The Bayesian information criteria (BIC) curve in the DAPC analyses supported K=6 with a clear discrimination of genotypes from China, Pakistan, Nepal, Middle East-East Africa, North western (NW) Europe and Central Asia-Mediterranean region (Fig. 1 and Fig. 2).

Analyses using the model-based clustering method implemented in STRUCTURE also identified an optimal number of clusters (K)equal to 6, based on the rate of change in the log probability of data across successive K values [59], and patterns of subdivision were largely consistent with the results of non-parametric DAPC (Fig. 2 and Fig. S2). At K=2, Middle Eastern, Mediterranean and Central Asian populations were assigned to one group; the Chinese population was assigned to the other group; and Nepalese, Pakistani and NW European populations had a mixed assignment of the two groups (Fig. S2). Increasing K to 3 individualized a Pakistan-specific group, while increasing K to 4 split the cluster of Middle East, Central Asia and Mediterranean region into two groups, one specific to the Middle East and East Africa and the other specific to the Central-Asia and Mediterranean region, with substantial admixture from the Middle East. The Middle Eastern and East African populations had no differentiation from each other and are termed as Middle East-East Africa, onward. At K = 5, the NW European populations were separated from the Chinese population, and at K=6, the Nepalese group individualized (Fig. S2). Increasing K above 6 did not reveal any further subdivisions (Fig. S2).

Population differentiation among the different geographically spaced populations was estimated by means of pairwise F_{st} . Populations showed a strong differentiation, with high and significant F_{st} values for all pairs except for PST from the Middle Eastern, Central Asian and Mediterranean regions (Table 2), indicative of a relatively recent shared ancestry or significant gene flow among these three latter populations. Chinese, Pakistani and Nepalese populations were differentiated from one another and from the Middle Eastern and Mediterranean populations. These two latter populations were not highly differentiated from one another (Fig. S2; Table 2). The NW European population showed a strong differentiation from Mediterranean and Middle Eastern populations but was closer to the Chinese population (Fig. S2 and Fig. S3).

Geographical patterns of genetic and genotypic variability

Populations from NW Europe, North America, South America, Australia, South Africa, East Africa, the Middle East and the



Figure 2. Clustering of 409 PST isolates representing worldwide geographical regions to genetic groups for the optimal K-value (K = 6) in the DAPC analysis. PstS1 and PstS2 refers to the two closely related aggressive strains, while PstS3 refers to the older aggressive isolates regularly reported in Southern Europe. doi:10.1371/journal.ppat.1003903.g002

Pape Z. Estimates of F _{ST} (i populations.	upper diagor	ind signi	ricance (lower d	lagonal) based on 20 ff		1107 085 101 1	solates repre	senting world	wide geo	ographically	spaced
	NW Europe	North America	South America	Mediterranean Region	Central Asia	South Africa	East Africa	Middle East	Nepal	Pakistan	China
NW Europe		0.039	0.001	0.420	0.380	0.498	0.500	0.380	0.370	0.410	0.390
North America	0.100	I	0.046	0.409	0.368	0.485	0.490	0.378	0.364	0.400	0.398
South America	0.410	0.100	1	0.435	0.396	0.514	0.511	0.393	0.379	0.416	0.405
Mediterranean Region	0.000	0.000	0.000		0.020	0.109	0.150	0.009	0.280	0.280	0.390
Central Asia	0.000	0.000	0.000	0.000	ı	0.044	0.160	0.040	0.230	0.260	0.340
South Africa	0.000	0.000	0.000	0.010	0.190	ı	0.229	0.160	0.298	0.314	0.419
East Africa	0.000	0.000	0.000	0.000	0.000	0.600	ı	0.140	0.380	0.280	0.540
Middle-East	0.000	0.000	0.000	0.020	0.000	0.000	0.000		0.260	0.250	0.360
Nepal	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000		0.220	0.210
Pakistan	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000		0.450
China	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	,
F_{ST} for aggressive strains	0.420	0.408	0.434	0.000	0.010	0.099	0.150	0.010	0.270	0.270	0.390
P-value for aggressive strains	0.000	0.000	0.000	0.860	0.070	0.040	0.000	0.020	0.000	0.000	0.000
The lower two lines shows F _{5T} and doi:10.1371/journal.ppat.1003903.tt	ł its p-value for i 302	isolates representing	the post-2000 eme	rged strains. Non-significant	F_{ST} values (>0.01)	are shown in bolc					

Worldwide Wheat Yellow Rust Population Structure

Mediterranean region displayed low genotypic diversity as well as an excess of heterozygosity compared to expectations under HWE, consistent with long-term clonality. Asia appeared as the zone of the highest diversity of the pathogen, with Himalayan (Nepalese and Pakistani) and near Himalayan (Chinese) populations not departing from HWE, suggesting the occurrence of recombination within the populations (Fig. 3), and showing higher variability in terms of genotypic diversity and allele richness (Fig. 4). Populations from Middle East and East Africa were more diverse than the European and Mediterranean populations, where the maximum clonal resampling was observed. Pakistan displayed the highest number of private alleles (Fig. 4). Isolates representing NW Europe also had high private allele richness, probably due to their strict clonality [19,25] and isolation from other populations.

Source of recently emerged populations

We detected only a few recent migrants, admixed and unassigned isolates in each geographical region in clustering analyses (Fig. 2 and Fig. S2). The existence of such genotypes in the Himalayan and neighboring regions may reflect back migrations or shared ancestral variation, as both phenomena give the same signal with clustering algorithms [60,61]. Clear migration footprints were only found when focusing on recently colonized areas. Analyses confirmed NW Europe as the source of the North American and Australian populations, and the Mediterranean region and Central Asia appeared to be the source of the South African population (Fig. 2 and Fig. S2). Additionally, the South American isolates were assigned to NW European isolates and displayed very low diversity, revealing another incursion from NW Europe.

Isolates of the recently emerged aggressive strain PstS1, associated with the post-2000 epidemics in the USA and Australia, consisted of only a single multilocus genotype (MLG-99) resampled in other geographical regions as well. PstS1 was closely related to the other recently emerged aggressive strain reported in Asia, Africa and Europe, the PstS2, which consisted of different multilocus genotypes, including this MLG-99. Both PstS1 and PstS2 were assigned to the Middle Eastern-East Africa group, suggesting a source in Middle East-East Africa for these strains (Fig. 2 and Table 2). An older set of aggressive isolates frequently reported in Southern Europe [25], although with lesser number of virulences than PstS1 and PstS2 [24], were assigned to the Central Asian-Mediterranean genetic group (represented as PstS3 in Fig. 2).

Ancestral relationship and migration patterns of populations

We used approximate Bayesian computation (ABC) analyses to infer on the ancestral relationship among populations. To limit the number and complexity of the scenarios to be compared, we used a sequential approach: we defined nested subsets of competing scenarios based on our understanding of the population structure in different regions, and analyzed these subsets sequentially. The origin of recently emerged populations was investigated using clustering and differentiation analyses and it has been presented in the previous section.

We started by analyzing the historical relationships among the three populations displaying recombining population structures and located in the region of highest diversity (i.e., Pakistan, Nepal and China). The four competing scenarios assumed three different population-trees, and admixture in the ancestry of the Nepalese population (Fig. S4) as this population appeared admixed in population subdivision analyses (Fig. S2 and Fig. S3). Leave-one-out cross-validation for model selection [52] confirmed that our



Figure 3. Expected (He) and observed (Ho) heterozygosity for clone-corrected data based on 20 polymorphic microsatellite loci for PST isolates sampled from diverse geographical regions.

doi:10.1371/journal.ppat.1003903.g003

methodology was able to distinguish between the four different scenarios (Table S5). The emergence of the Nepalese populations following admixture between Pakistani and Chinese populations appeared the most likely [Scenario 4a, posterior probability (PP): 0.9911; (Table S6)]. This scenario (4a) was then used as a



Geographical Origin

backbone to design scenarios investigating the origin of other populations. Because a relatively recent origin of the Nepalese population was indicated by parameter estimates [Maximum Posterior Probability (MPP) estimate: 103 generations; 95% CI: 14–332; (Table S3)], this population was therefore dismissed as a possible source of other populations and not considered in subsequent analyses.

The origin of Central Asian, Middle Eastern and Mediterranean population was investigated jointly, based on the relatedness of these populations in population structure analyses (Fig. 1, S2 and Table 2), and their geographical proximity from the centre of diversity (the Himalayan and neighboring areas). Central Asian and the Mediterranean populations were pooled, based on their genetic relatedness to limit the number of competing scenarios. Twelve scenarios were compared, assuming two different native populations (China and Pakistan), sequential or independent introductions and admixture (Fig. S5). Leave-one-out analyses indicated a good ability to distinguish between the twelve scenarios (Table S5). Scenario 11b, assuming that the Central Asian, Middle Eastern and Mediterranean populations split following an admixture event between Chinese and Pakistani populations appeared the most likely (PP = 0.717; Table S6; Fig. 5). The Pakistani population had a slightly higher contribution to the admixture event [1-r5=0.57, 95% Credibility interval (CI): 0.096-0.954] than the Chinese population, which might account for the assignment of simulated dataset under this scenario to scenario 10b (both populations originated from Pakistan) in leaveone-out analyses. We also considered the possibility that populations not represented in our dataset could have contributed to the

Figure 4. Diversity index, allele richness (A) and private allele richness (B) for PST populations from worldwide geographical regions. doi:10.1371/journal.ppat.1003903.q004

genetic makeup of extant worldwide PST populations. All scenarios including a hypothetical un-sampled source population had lower posterior probabilities than the most supported scenario (11b) described above (scenarios set d; Fig. S7; Table S6). Leaveone-out analyses, however, indicated limited power to distinguish among the scenarios (Table S5).

The most likely scenario identified in analyses of Central Asian, Middle Eastern and Mediterranean populations was then used as a backbone to investigate the origin of the NW European population. Six scenarios were compared, assuming three different populations of origin (China, Pakistan and the Central-Asian-Middle-Eastern-Mediterranean lineage identified above) and admixture (Fig. S6). The scenario (2c) assuming that the NW European emerged following an admixture event between Chinese and Pakistani populations appeared the most likely (PP = 0.499); Table S6; Fig. 5). Parameter estimates for scenario 2c suggested a higher contribution of the Chinese population to the admixture event (1-r7 = 0.756, 95% CI: 0.187-0.987), which may account for the relatively high posterior probabilities for scenarios 1c and 4c [PP=0.173 and PP=0.270, respectively]. Inclusion of a hypothetical un-sampled source population revealed a slightly higher posterior probability of scenario 2c described above (PP = 0.303) than the posterior probability of scenario 4e (PP = 0.287) assuming that the NW European population resulted from an admixture event between the Chinese and un-sampled population (scenarios set e; Fig.S7; Table S6). Like scenario 2c, parameter estimates for scenario 4e suggested a higher contribution of the Chinese population to the admixture event (r8=0.797, 95%CI: 0.069-0.982).

Discussion

We investigated the origin, migration routes and population structure of wheat yellow rust pathogen *Puccinia striiformis* f.sp. *tritici* (PST) using a comprehensive set of isolates from six continents and 20 highly informative SSR markers. We showed that despite the long-distance dispersal and the recent global spread of aggressive strains, the worldwide PST population has a clear genetic structure and is separated into six groups corresponding to main epidemic areas. The maintenance of a clear genetic structure despite substantial gene flow results from both strong clonality in many PST populations, and strong founder effects in recently emerged populations.

Strong population subdivision despite long-distance dispersal

We report the existence of a strong population subdivision within PST, with a clustering of isolates according to their geographical origin despite a capacity for long-distance migration [9,11]. This pattern stands in stark contrast with the previous understanding of the worldwide population structure of PST, which considers the potential replacement of local populations by new invasions [5,9]. On the basis of pathological survey that monitor the occurrence of strains with newly acquired virulences that defeat recently deployed Υr (resistance) genes, the population structure of PST was considered to be shaped by a continual replacement of pre-existing populations by the newly emerged and spread pathotypes, or aggressive strains. This process is well known as the boom and bust cycle [5,10]. However, such surveys were designed to track the spread of a new, virulent race and, therefore, were potentially biased due to sampling only from varieties with the defeated resistance gene(s) in question but not from local landraces or other varieties. These observations lead to Asia (except China) being considered as a single epidemiological zone,

with rapid and recurrent spread of new virulences over the whole zone, as in the case of virulence matching the $\Upsilon r9$ resistance gene (Fig. 6 [5]) and the recent virulence matching the stem rust resistance gene Sr31 [4]. Indeed, such geographic migrations are also documented in our study, but recently spread genotypes appear to coexist with and are dominated by older populations specific to the main geographic areas, suggesting that migrants do not replace local populations in recombinant Asian populations despite the capacity for recurrent and long-distance dispersal. In contrast, the invasion of new genotypes in clonal populations would result in a population sweep and would replace the original population. This was observed in the USA, where the post-2000 PST population is dominated by the pathotypes characteristic of the aggressive strain, PstS1, or its derivatives, shown above to have originated from the East African-Middle Eastern region. The outcome of new pathotype introductions may depend on the relative competitiveness of local populations in their region of origin, and the selective advantage of migrants (e.g. their virulence towards a new resistance gene widely deployed or an increased tolerance to prevalent abiotic stresses e.g., high temperatures).

Regional differences in levels of recombination

PST has long been considered a strictly asexual pathogen on wheat due to the lack of knowledge about the existence of alternate host to complete sexual reproduction [16,62]. Population genetic surveys that revealed clonal populations with very low diversity in USA [27], Europe [25] and Australia [15] were consistent with a hypothesis of asexual reproduction. Recently, populations with higher diversity were reported in the Middle East [31] and Pakistan [29], and a recombinant population structure was evidenced in China [28]. Herein, we identified a recombinant population structure and high diversity in Nepal and Pakistan and confirmed previous findings in China, suggesting the existence of possible sexual reproduction in PST populations from a broad area ranging from the Himalayan region to the Mongolian plateau. This possibility also recently gained indirect experimental support, with the demonstration of Berberis spp. as alternate host for PST under laboratory conditions [63] and a high ability for sexual reproduction (in the form of telial production) reported in the Asian populations of PST [32]. Although the role of Berberis spp. for the life cycle of PST under natural conditions remains to be further investigated, the presence of *Berberis spp.* in Pakistan, Nepal and China [64,65,66] is consistent with the existence of a sexual cycle of the pathogen in Asia.

On the source of new incursions and emergences

Clustering and differentiation analyses allowed us to identify the source of new incursions and emergence (Fig. 2 and Fig. 6). The source of the Australian and North American populations was confirmed to be NW Europe, in accordance with previous findings, suggesting the migration of PST from NW Europe to Australia in 1979 [15] and probably earlier to North America [11,67]. We also identified the NW European source of the South American PST population, which was reported earlier in the 20th century with no inference on its source [16,21]. This suggests that PST incursions into both North and South America originated from NW Europe, probably through human intervention. We also identified the Mediterranean-Central Asian population as the source of South African populations, first reported in 1996 [20], which might have resulted from wind dispersal or human intervention (Fig. 6).

Two closely related strains of PST, distinct from local populations, were recently reported in North America, Australia and Europe [6]. These strains were shown to be highly aggressive



Figure 5. Ancestral relationship among worldwide PST populations as inferred from the analyses of Approximate Bayesian Computations. doi:10.1371/journal.ppat.1003903.q005

and adapted to high temperature [24]. One of the strains (PstS1) was responsible for PST epidemics in south-central USA, a region previously considered too warm for yellow rust epidemics [in 2000; 22,68], and in Western Australia [in 2002; 69]. Another strain (PstS2), closely related to the first one, was reported in NW Europe with similar aggressiveness and strong differentiation from local PST populations. PstS2 was also shown to be present in the Mediterranean region and the Middle East-East Africa [6]. Our analyses revealed that PstS1 representative isolates had a single multilocus genotype (MLG-99), while PstS2 consisted of different, but closely related, MLGs. Assignment analyses revealed that both strains originated from the Middle East-East Africa, and such global patterns of dispersal would involve accidental spore transport linked with human activities. Thus, the incursion of PST into the Americas and the spread of aggressive strains are

most probably the direct consequence of human-associated dispersal, as suggested for the initial introduction of PST in Australia [15]. These results suggests that PST may spread rapidly through winds or human activities, with former playing a greater role at regional scales [26], the latter could be involved in inter-continental spread.

Origin of the worldwide PST populations

Transcaucasia had previously been suggested as the centre of origin for the pathogen, mainly based on its diversity of virulence and distribution of pathotypes [16,17]. However, the diversity of virulence and the distribution of pathotypes are strongly influenced by the resistance in host populations, and this might lead to biased inferences of the location of the centre of origin of the pathogen. In our analyses, the representative isolates from



Figure 6. Origin and migration routes of recently emerged populations of wheat yellow rust pathogen identified or confirmed through population genetic analyses of a worldwide representative set of isolates. The year shows the first report of incidence based on present work and previous work (Aggressive strain = Hovmøller et al., 2011; South African population = Boshoff et al., 2002; Spread of Vir9 = Singh et al., 2004; incursion into Australia = Wellings and McIntosh, 1990). doi:10.1371/journal.ppat.1003903.q006

Transcaucasia were less diverse, appeared to be clonal and did not exhibit strong divergence from the rest of the Middle-Eastern, Central Asian and Mediterranean populations. In contrast, the existence of high levels of diversity, private alleles, a recombinant population structure, the ability to produce sex-related structures [32] and the independent maintenance of PST populations in the Himalaya [66] suggest this latter region as a more plausible centre of origin for PST. The analysis of ancestral relationships among worldwide populations further confirmed the Himalayan populations to be the ancestral populations and the differentiation between Pakistan and China to be the most ancestral split (Fig. 5; Table S2, S3, S4, S5, S6). If one considers that the centre of origin of PST is in the Himalayan region, then PST would have adapted to wheat through host-range expansion or host shift and not followed a host-tracking co-evolution with early wheat domestication in the Fertile Crescent. This would add to the increasingly adopted view that infection of a novel host is a major route of disease emergence in fungal pathogens [70]. Additional sampling campaigns and full-genome re-sequencing of new isolates from wild and domesticated hosts should provide further insight into the history of PST and advance our understanding of the evolutionary response of pathogens to plant domestication and the development of agro-systems.

Proposed scenario for historical migration routes of PST

Once adapted to wheat in its centre of origin in the Himalayas or the nearby regions, PST could have spread to the rest of the world while evolving independently in different parts of the Himalayan region, resulting in population subdivision within the native area [66]. PST could have spread northward from the Himalayas to the Mongolian plateau in China, where it maintained sexual reproduction and a high diversity in some parts, with an acquisition of virulences to the wheat varieties grown in China. Eastbound, ABC analyses indicated that relatively recent admixture event between Pakistani and Chinese populations resulted in the Nepali population. On the westward side of the Himalayan region, the three Middle-Eastern, Mediterranean and Central Asian populations could be the result of an admixture between Pakistan and China (Fig. 5) with relatively higher contribution of the Pakistani population to the admixture event according to ABC inferences. The Mediterranean, Middle Eastern and Central Asian populations were less differentiated, despite their distribution across a large geographical area. The offseason maintenance in some regions, its subsequent spread to other Middle-Eastern, Mediterranean and Central Asian regions and the lack of local off-season survival through volunteers or sexual reproduction in all regions could result in a source and sink relationship at the scale of this whole region, as previously suggested [31]. The most recent incursion from this Middle-Eastern, Mediterranean and Central Asian population was into South Africa, where PST was absent until 1996 [20].

The NW European population was shown in ABC analyses to be the result of admixture between China and Pakistan or between China an un-sampled source population with a higher contribution of the Chinese population to the admixture event. This is consistent with the lower differentiation between the NW European population and the Chinese population (Table 2), compared to the Pakistani or the Middle Eastern, Central Asian and Mediterranean populations. These results suggest that PST has spread from China to NW Europe, and this spread more likely occurred through some human intervention rather than an airborne incursion from the Middle Eastern, Central Asian and/ or Mediterranean regions. This NW European population succeeded in terms of off-season survival on volunteers in coastal areas with mild winter and resulted in a reduced ability for sexual reproduction [32]. The clonal evolution within the NW European population resulted in a strong negative $F_{\rm IS}$ value, owing to a clonal divergence of the dikaryotic genomes [25], in line with what is expected according to the Meselson Effect [71]. From NW Europe, PST has subsequently been introduced to North and South America and even more recently to Australia (Fig. 6).

Our study suggests the Himalayan region as the likely centre of origin of PST, it confirms the previous migration hypotheses (invasion of USA and Australia from NW Europe and the source of aggressive strains), and it also provides an integrative scenario for the worldwide spread of PST including new findings regarding the origins of South American and European populations (Fig. 6). More extended surveys, particularly in northern parts of Asia and Eastern part of Himalayan region along with a multigene phylogeny approach based on several gene sequences will provide a more detailed resolution in the analysis of the colonization history of the pathogen and its population structure in Asia, taking advantage of the historical/epidemiological records of PST emergences.

Conclusions

The existence of a high genotypic diversity, a high ability for sexual reproduction as well as the independent maintenance of strongly differentiated populations in the Himalayan region suggests this region as the putative centre of origin of PST. Differences in the levels of diversity and mode of reproduction among geographically distant populations are particularly relevant in the context of risk-assessment for disease emergence: Asian populations (China, Nepal and Pakistan) with a high level of recombination, diversity and ability for sexual reproduction could serve as possible sources for the emergence of new, virulent and aggressive strains. The maintenance of populations specific to geographical regions in Asia suggests a survival of local populations in these regions, potentially through sexual reproduction. For integrated disease management, it would be important to quantify the relative contribution of sexual vs. asexual reproduction to the diversity in different world populations and identify sexual host(s) or clonal over-summering/-wintering pathways. Finally, this study emphasizes the potential role of human travel and commerce as a major driver in the emergence of PST. The intensification of business and tourism activities in regions known as major sources of pathogen diversity should be considered in the context of risks associated with the emergence and worldwide spread of plant disease.

Supporting Information

Figure S1 Box-plot of number of PST genotypes detected as a function of the number of loci re-sampled 1000 times within the 20 microsatellite markers (from 0–20) using GENCLONE software. The box represents the average, minimum and maximum numbers of MLGs detected when re-sampling on loci. (DOC)

Figure S2 Assignment of PST isolates from worldwide geographical regions to genetic groups, using the STRUCTURE software and different K-values (genetic groups). The chart represents the consensus assignment obtained by analysis of result from 20 runs of the STRUCTURE analysis with the CLUMPP software. Each color represents a different genetic group. (DOC)

Figure S3 Microsatellite distance-based Neighbor-Joining tree for PST isolates from worldwide geographically spaced

populations. Agr = Aggressive strain, CA = Central Asia, Ch = China, Erit = East Africa, MdlEst = Middle East, Med = Mediterranean region, Nep = Nepal, NWEu = NW European group, Pk = Pakistan and SA = South Africa. (DOC)

Figure S4 Scenarios regarding the evolutionary relationship among the three recombinant populations from the centre of diversity of *Puccinia striiformis* f.sp. *tritici.* (DOC)

Figure S5 Scenarios regarding the evolutionary relationship between the two ancestral populations of Pakistan and China with the Middle Eastern and Central Asian-Mediterranean populations.

(DOC)

Figure S6 Scenarios regarding the evolutionary relationship between the two ancestral populations of Pakistan and China with Middle Eastern, Central Asian-Mediterranean and NW European populations.

(DOC)

Figure S7 Scenarios regarding the evolutionary relationship between the two ancestral populations of Pakistan and China with Middle Eastern, Central Asian-Mediterranean and NW European populations along with a 'ghost' population. (DOC)

Table S1Gene diversity for 20 SSR loci in PST populations ofdiverse geographical origin.

(XLS)

 Table S2
 Prior distributions of model parameters.

 (XLS)
 (XLS)

Table S3 Demographic and mutation parameter estimated using Approximate Bayesian Computation. Only parameter esimtates for the scenarios having highest posterior probabilities are shown. Demographic parameters are introduced in corresponding Figures. Composite parameters scaled by the mutation rate are also shown. The mutation parameters are μ (mean

References

- Stukenbrock EH, McDonald BA (2008) The origins of plant pathogens in agrocosystems. Annual Review of Phytopathology 46: 75–100.
- Parker IM, Gilbert GS (2004) The evolutionary ecology of novel plant-pathogen interactions. Annual Review of Ecology, Evolution, and Systematics 35: 675– 700.
- Palm M (2001) Systematics and the impact of invasive fungi on agriculture in the United States. BioScience 51: 141–147.
- Singh RP, Hodson DP, Jin Y, Huerta-Espino J, Kinyua MG, et al. (2006) Current status, likely migration and strategies to mitigate the threat to wheat production from race Ug99 (TTKS) of stem rust pathogen. CAB Reviews 1: 054.
- Singh RP, William HM, Huerta-Espino J, Rosewarne G (2004) Wheat rust in Asia: meeting the challenges with old and new technologies. In proceedings of the 4th International Crop Science Congress; 26 Sep–1 Oct 2004; Brisbane, Australia. Available at: http://www.cropscience.org.au/icsc2004/symposia/3/ 7/141_singhtp.htm.
- Hovmoller MS, Yahyaoui AH, Milus EA, Justesen AF (2008) Rapid global spread of two aggressive strains of a wheat rust fungus. Molecular Ecology 17: 3818–3826.
- Perrings C, Williamson M, Barbier E, Delfino D, Dalmazzone S, et al. (2002) Biological invasion risks and the public good: an economic perspective. Conservation Ecology 6: Article 1.
- Campbell F (2001) The science of risk assessment for phytosanitary regulation and the impact of changing trade regulations. BioScience 51: 148–153.
- Brown JKM, Hovmøller MS (2002) Aerial dispersal of fungi on the global and continental scales and its consequences for plant disease. Science 297: 537–541.
- de Vallavicille-Pope C, Ali S, Leconte M, Enjalbert J, Delos M, et al. (2012) Virulence dynamics and regional structuring of *Puccinia striiformis* f. sp. tritici in France between 1984 and 2009. Plant Disease 96: 131–140.
- Hovmøller MS, Sørensen CK, Walter S, Justesen AF (2011) Diversity of Puccinia striiformis on cereals and grasses. Annual Review of Phytopathology 49: 197–217.

mutation rate) and p (mean value of the geometric distribution parameter that governs the number of repeated motifs that increase or decrease the length of the locus during mutation events).

(XLS)

Table S4Model checking for demographic scenarios analysedusing Approximate Bayesian Computation.(XLS)

Table S5 Confidence in model choice for the demographic scenarios compared using approximate Bayesian computations. Confidence in model choice was assessed using a leave-one-out method. For each scenario 100 of the 500,000 simulated datasets used for model selection were randomly drawn and treated as pseudo-observed datasets. Posterior probabilities of competing scenarios were evaluated for each pseudo-observed dataset, using all remaining simulated datasets. Confidence in model choice was then estimated as the proportions of simulated data assigned to each scenario.

(XLS)

Table S6 Posterior probabilities (PP) of the different sets of scenarios compared using Approximate Bayesian Computation. (XLS)

Acknowledgments

The authors are grateful to all their colleagues who participated in establishing the yellow rust collection used for this study. We also thank Laurent Gérard, Nathalie Galet, Ellen Frederiksen and Vibeke Mikkelsen for their technical assistance during the multiplication of yellow rust isolates and DNA extraction. Finally, we would like to acknowledge Stephanie Walter for her constructive comments on the manuscript, and the Associate Editor and anonymous reviewers for many useful suggestions.

Author Contributions

Conceived and designed the experiments: MSH JE CdVP. Performed the experiments: SA. Analyzed the data: SA PG. Contributed reagents/ materials/analysis tools: ML AG AFJ. Wrote the paper: SA PG JE.

- Hovmøller MS, Walter S, Justesen AF (2010) Escalating threat of wheat rusts. Science 329: 369.
- Ali S, Shah SJA, Rahman H, Saqib MS, Ibrahim M, et al. (2009) Variability in wheat yield under yellow rust pressure in Pakistan. Turkish Journal of Agriculture and Forestry 33: 537–546.
- Ali S, Shah SJA, Khalil IH, Rahman H, Maqbool K, et al. (2009) Partial resistance to yellow rust in introduced winter wheat germplasm at the north of Pakistan. Australian Journal of Crop Science 3: 37–43.
- Wellings CR (2007) Puccinia striiformis in Australia: A review of the incursion, evolution and adaptation of stripe rust in the period 1979–2006. Australian Journal of Agricultural Research 58: 567–575.
- Stubbs RW (1985) Stripe rust. In: Roelfs AP, Bushnell WR, editors. The Cereal Rusts, Diseases, Distribution, Epidemiology and Control. London, UK: Academic Press. pp. 61–101.
- Stubbs RW (1988) Pathogenicity analysis of yellow (stripe) rust of wheat and its significance in a global context. In: Simmonds NW, Rajaram S, editors. Breeding strategies for resistance to the rusts of wheat: CIMMYT D.F. Mexico.
- Kolmer JA (2005) Tracking wheat rust on a continental scale. Current Opinion in Plant Biology 8: 441–449.
- Justesen AF, Ridoutb CJ, Hovmøller MS (2002) The recent history of *Puccinia striiformis* f.sp. tritici in Denmark as revealed by disease incidence and AFLP markers. Plant Pathology 51: 13–23.
- Boshoff WHP, Pretorius ZA, van Niekerk BD (2002) Establishment, distribution, and pathogenicity of *Puccinia striiformis* f.sp. *tritici* in South Africa. Plant Disease 86: 485–492.
- Rudorf W, Job M (1934) Untersuchungen bezüglich des Spezialisierung von Puccinia graminis tritici. Puccinia triticina und Puccinia glumarum tritici, sowie über Resistenz und ihre Vererbung in verschiedenen Kreuzungen. Pflanzenzüchtung 19: 333–365.
- Chen XM (2005) Epidemiology and control of stripe rust [Puccinia strijformis f: sp. tritici] on wheat. Canadian Journal of Plant Pathology 27: 314–337.

- Markell SG, Milus EA (2008) Emergence of a Novel Population of *Puccinia* striiformis f. sp. tritici in Eastern United States. Phytopathology 98: 632–639.
- Milus EA, Kristensen K, Hovmøller MS (2009) Evidence for increased aggressiveness in a recent widespread strain of *Puccinia striiformis* f.sp. tritici causing stripe rust of wheat. Phytopathology 99: 89–94.
- Enjalbert J, Duan X, Leconte M, Hovmøller MS, de Vallavieille-Pope C (2005) Genetic evidence of local adaptation of wheat yellow rust (*Puccinia striiformis* f. sp. *tritici*) within France. Molecular Ecology 14: 2065–2073.
- Hovmøller MS, Justesen AF, Brown JKM (2002) Clonality and long-distance migration of *Puccinia striiformis* f.sp. *tritici* in north-west Europe. Plant Pathology 51: 24–32.
- Chen X, Line RF, Leung H (1993) Relationship between virulence variation and DNA polymorphism in *Puccinia striiformis*. Phytopathology 83: 1489–1497.
- Mboup M, Leconte M, Gautier A, Wan AM, Chen W, et al. (2009) Evidence of genetic recombination in wheat yellow rust populations of a Chinese oversummering area. Fungal Genetics and Biology 46: 299–307.
- Bahri B, Shah SJA, Hussain S, Leconte M, Enjalbert J, et al. (2011) Genetic diversity of wheat yellow rust population in Pakistan and its relationship with host resistance. Plant Pathology 60: 649–660.
- Hovmøller MS, Justesen AF (2007) Appearance of atypical Puccinia striiformis f. sp. tritici phenotypes in north-western Europe. Australian Journal of Agricultural Research 58: 518–524.
- Bahri B, Leconte M, Ouffroukh A, de Vallavieille-Pope C, Enjalbert J (2009) Geographic limits of a clonal population of wheat yellow rust in the Mediterranean region. Molecular Ecology 18: 4165–4179.
- Ali S, Leconte M, Walker A-S, Enjalbert J, de Vallavieille-Pope C (2010) Reduction in the sex ability of worldwide clonal populations of *Puccinia striiformis* f.sp. tritici. Fungal Genetics and Biology 47: 828–838.
- Enjalbert J, Duan X, Giraud T, Vautrin C, de Vallavieille-Pope C, et al. (2002) Isolation of twelve microsatellite loci, using an enrichment protocol, in the phytopathogenic fungus *Puccinia striiformis* f. sp. *tritici*. Molecular Ecology Notes 2: 563–565.
- 34. Ali S, Gautier A, Leconte M, Enjalbert J, De Vallavieille-Pope C (2011) A rapid genotyping method for an obligate fungal pathogen, *Puccinia striifornis* f.sp. *tritici*, based on DNA extraction from infected leaf and Multiplex PCR genotyping. BMC Research Notes 4: 240.
- Dutech C, Fabreguettes O, Capdevielle X, Robin C (2010) Multiple introductions of divergent genetic lineages in an invasive fungal pathogen, *Cryphonectria parasitica*, in France. Heredity 105: 220–228.
- Jombart T, Devillard S, Balloux F (2010) Discriminant analysis of principal components: a new method for the analysis of genetically structured populations. BMC Genetics 11: 95 http://www.biomedcentral.com/1471-2156/1411/1494.
- Pritchard JK, Stephens M, Donnelly P (2000) Inference of population structure using multilocus genotype data. Genetics 155: 945–959.
- Gao H, Williamson S, Bustamante CD (2007) A markov chain monte carlo approach for joint inference of population structure and inbreeding rates from multilocus genotype data. Genetics 176: 1635–1651.
- Jakobsson M, Rosenberg NA (2007) CLUMPP: a cluster matching and permutation program for dealing with label switching and multimodality in analysis of population structure. Bioinformatics 23: 1801–1806.
- Nei M, Tajima F, Tateno Y (1983) Accuracy of estimated phylogenetic trees from molecular data. Journal of Molecular Evolution 19: 153–170.
- Langella O (2008) POPULATIONS 1.2.3, Logiciel de génétique des populations. In: Langella O, editor. Gif-sur-Yvette, France: Laboratoire Populations, Génétique et évolution, CNRS.
- 42. Belkhir KBP, Chikhi L, Raufaste N, Bonhomme F (2004) GENETIX 4.05, logiciel sous Windows TM pour la génétique des populations. Laboratoire Génome, Populations, Interactions, CNRS UMR 5000: Université de Montpellier II, Montpellier (France) http://www.univ-montp2.fr/~genetix/ genetix.htm.
- Arnaud-Haond S, Belkhir K (2007) GenClone 1.0: A new program to analyse genetics data on clonal organisms. Molecular Ecology Notes 7: 15–17.
- Goudet J (2001) FSTAT, a program to estimate and test gene diversities and fixation indices.2.9.3 ed: Available from http://www2.unil.ch/popgen/ softwares/fstat.htm.
- Agapow PM, Burt A (2001) Indices of multilocus linkage disequilibrium. Molecular Ecology Notes 1: 101–102.

- Szpiech ZA, Jakobsson M, Rosenberg NA (2008) ADZE: a rarefaction approach for counting alleles private to combinations of populations. Bioinformatics 24: 2498–2504.
- Raymond M, Rousset F (1995) GENEPOP Version 1.2: population genetics software for exact tests and ecumenicism. Journal of Heredity 86: 248–249.
- Maynard-Smith J, Smith NH, O'rourke M, Spratt BG. How clonal are bacteria?; 1993. pp. 4384–4388.
- Dilmaghani A, Gladieux P, Gout L, Giraud T, Brunner PC, et al. (2012) Migration patterns and changes in population biology associated with the worldwide spread of the oilseed rape pathogen *Leptosphaeria maculans*. Molecular Ecology 21: doi: 10.1111/j.1365-294X.2012.05535.x.
- Cornille A, Gladieux P, Smulders MJM, Roldan-Ruiz I, Laurens F, et al. (2012) New Insight into the History of Domesticated Apple: Secondary Contribution of the European Wild Apple to the Genome of Cultivated Varieties. PLoS Genetics 8: e1002703. doi:1002710.1001371/journal.pgen.1002703.
- Cornuet JM, Ravigne V, Estoup A (2010) Inference on population history and model checking using DNA sequence and microsatellite data with the software DIYABC (v1.0). BMC Bioinformatics 11: 401.
- Csilléry K, François O, Blum M (2012) abc: an R package for approximate Bayesian computation (ABC). Methods in Ecology and Evolution 3: 475–479.
- Nei M (1978) Estimation of average heterozygosity and genetic distance from a small number of individuals. Genetics 89: 583–590.
- Weir BS, Cockerham CC (1984) Estimating F-statistics for the analysis of population structure. Evolution Psychiatrique 38: 1358–1370.
- Goldstein DB, Ruiz LA, Cavalli-Sforza LL, Feldman MW (1995) An evaluation of genetic distances for use with microsatellite loci. Genetics 139: 463–471.
- Fagundes NJR, Ray N, Beaumont MA, Neuenschwander S, Salzano FM, et al. (2007) Statistical evaluation of alternative models of human evolution. Proceedings of the National Academy of Science of the United States of America 104: 17614–17619.
- Cornuet JM, Santos F, Beaumont MA, Robert CP, Marin JM, et al. (2008) Inferring population history with DIY ABC: a user-friendly approach to approximate Bayesian computation. Bioinformatics 24: 2713–2719.
- Beaumont MA, Zhang W, Balding DJ (2002) Approximate Bayesian computation in population genetics. Genetics 162: 2025–2035.
- Evanno G, Regnaut S, Goudet J (2005) Detecting the number of clusters of individuals using the software STRUCTURE : a simulation study. Molecular Ecology 14: 2611–2620.
- Gladieux P, Zhang X-G, Afoufa-Bastien D, Sanhueza R-MV, Sbaghi M, et al. (2008) On the origin and spread of the scab disease of apple: out of Central Asia. PLoS ONE 3: e1455.
- Schofl G, Schlotterer C (2006) Microsatellite variation and differentiation in African and non-African populations of *Drosophila simulans*. Molecular Ecology 15: 3895–3905.
- Cummins GB (1971) The rusts fungi of cereals, grasses, and bamboos: Springer-Verlag, New York.
- Jin Y, Szabo LJ, Carson M (2010) Century-old mystery of Puccinia striiformis life history solved with the identification of Berberis as an Alternate Host. Phytopathology 100: 432–435.
- Ray R, Gururaja KV, Ramchandra TV (2011) Predictive distribution modeling for rare Himalayan medicinal plant *Berberis aristata* DC. Journal of Environmental Biology 32: 725–730.
- Perveen A, Qaiser M (2010) Pollen flora of Pakistan–LXV. berberidaceae. Pakistan Journal of Botany 42: 1–6.
- 66. Ali S, Gladieux P, Rahman H, Saqib MS, Fiaz M, et al. (2014) Inferring the contribution of sexual reproduction, migration and off-season survival to the temporal maintenance of microbial populations: a case study on the wheat fungal pathogen *Puccinia strijformis* f.sp. *tritici*. Molecular Ecology DOI: 10.1111/ mec.12629.
- Carleton MA (1915) A serious new wheat rust in this country. Science 42: 58–59.
 Milus EA, Seyran E, McNew R (2006) Aggressiveness of *Puccinia striiformis* f. sp. *tritici* isolates in the South-Central United States. Plant Disease 90: 847–852.
- Wellings CR, Wright DG, Keiper F, Loughman R (2003) First detection of wheat stripe rust in Western Australia: evidence for a foreign incursion. Australasian Plant Pathology 32: 321–322.
- Giraud T, Gladieux P, Gavrilets S (2010) Linking the emergence of fungal plant diseases with ecological speciation. Trends in Ecology & Evolution 25: 387–395.
- Bultin R (2002) The costs and benefits of sex: new insights from old asexual lineages. Nature Reviews Genetics 3: 313–317.